Author's response to reviews

Title: The expression of TRMT2A, a novel cell cycle regulated protein, identifies a subset of breast cancer patients with HER2 over-expression that are at an increased risk of recurrence

Authors:

David G Hicks (David_Hicks@urmc.rochester.edu)
Bagi R Janarthanan (Bagi.Janarthanan@roswellpark.org)
Ramya Vardarajan (Ramya.Vardarajan@RoswellPark.org)
Swati A Kulkarni (swati.kulkarni@roswellpark.org)
Thaer Khoury (Thaer.Khoury@roswellpark.org)
Daniel Dim (Daniel.Dim@roswellpark.org)
G. Thomas Budd (BUDDG@ccf.org)
Brian J Yoder (yoderbj@gmail.com)
Raymond Tubbs (tubbsr@ccf.org)
Marshall T Schreeder (mts@ccihsv.com)
Noel C Estopinal (NRMCT@aol.com)
Rodney A Beck (rbeck@applied-genomics.com)
Yanling Wang (ywang@applied-genomics.com)
Brian Z Ring (bzring@applied-genomics.com)
Robert S Seitz (rseitz@applied-genomics.com)
Douglas T Ross (dross@applied-genomics.com)

Version: 10 Date: 8 March 2010

Author's response to reviews:

Dear Editors,

One Final update, the lead author (David Hicks) has relocated to U. of Rochester. I have updated the .doc to reflect this.

Good to go!!!!
Doug